Author’s response to reviews

Title: Assessement of postoperative long-term survival quality and complications associated with radical antegrade modular pancreatosplenectomy and distal pancreatectomy: a Metameta-analysis and systematic review

Authors:
Zhou Quanyu (1652540230@qq.com)
Lei Zehua (leitsehua@126.com)
Gong Jie (827361656@qq.com)
Xie Qingyun (410321477@qq.com)
Liu Yu (243171791@qq.com)
Wang Qing (309696055@QQ.COM)
Zhao Xin (845489808@qq.com)

Version: 1 Date: 04 Aug 2018

Author’s response to reviews:

List of Responses

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled, “The comparison of radical antegrade modular pancreatosplenectomy versus distal pancreatectomy in complications and postoperative long term survival quality: a Meta analysis and systematic review” (BSUR-D-18-00306). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Response: Compared with the paper by Bruno Breda et al, we describe the Bruno Breda, who will laparoscopic distal pancreatectomy and robotic resection line systematic review. But we systematically evaluated the radical antegrade model of pancreatic splenectomy and distal pancreatectomy. These are two different comparison. In addition, we also compared patients' negligent complications and long-term quality of life. This is not done by Bruno Breda et al.

2. Response to comment: We did not include the complete PRISMA list as an attached file when submitting the revised manuscript.

Response: We have added the complete PRISMA list as an supplementary material.

3. Response to comment: the contributions of authors JG, QYX, QW, YL and XZ are not clear, and do not automatically qualify them for authorship.


4. Response to comment: provide figure titles/legends under a separate heading of 'Figure Legends' after the References.

Response: We have provided the title/illustration of the figure under the separate heading "Legend" after the reference.

5. Response to comment: Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file. Please also amend your figures, as Figure 1 in the text does not correspond to Figure 1 that is uploaded.

Response: Submit each manuscript as a single document and match the number.

Special thanks to you for your good comments.

Sincerely yours,

Zehua Lei, Dr
Professor and Chairman

Department of Hepatobiliary and Pancreatic Surgery

The Affiliated People's Hospital of Leshan City,

Leshan, Sichuan, 614000, PR China

e-mail: leitsehua@126.com

Tel: +86-13320921599